Dear Dr. Yumi Nakano,

Thank you for your attention to our manuscript. I have pleasure in sending you my revised manuscript.

These items are changed:

1. Page 7, line 20-21: The significant difference between BDI2 in the pregnant and non-pregnant subjects is added instead of the difference between BDI1 and BDI2.

2. Page 8, line 15: The analyses about BDI2 in pregnant and non-pregnant patients are added.

3. Page 11, line 12: In the sentence "We assume that the factors which affect BDI1 in the pregnant group and those in the non-pregnant group are the same, because Table 4 shows no statistically significant difference among mean score of BDI1 in the two groups." BDI1 was the correct phrase.
I do a typing error here (BDI1 is correct). Excuse me. If this part have a misinterpreting meaning, I will omit it.

4. Conclusion in Abstract: "weak" is corrected.

5. Page 8, line 19 and other lines: "multiple" is corrected.

Yours sincerely,

Dr. A. Khademi.